Author's response to reviews

Title: Epigenetic alterations of the keratin 13 gene in oral squamous cell carcinoma

Authors:

Kaori Naganuma (naganuma@college.fdcnet.ac.jp)
Mitsutoki Hatta (hatta@college.fdcnet.ac.jp)
Tetsuro Ikebe (ikb@college.fdcnet.ac.jp)
Jun Yamazaki (junyama@college.fdcnet.ac.jp)

Version: 2  Date: 11 August 2014

Author's response to reviews:

2014/08/11

Dear BMC Cancer Editorial Team,

Please find the revised version of manuscript #1517799865138998, entitled “Epigenetic alterations of the keratin 13 gene in oral squamous cell carcinoma”

We have made some changes to our manuscript according to the editorial suggestions.

1. Line numbering

The manuscript has been revised to include line and page numbers.

2. Ethics statement

An ethics statement has been added to the methods section.

“Ethics statement

All experiments in this manuscript have been approved by the Fukuoka Dental College Institutional Biosafety Committee.”

3. Abstract

The abstract section has been revised to be structured (Background, Methods, Results, and Conclusions) according to the guidelines for authors.

Sincerely,
Mitsutoki Hatta, D.D.S., Ph.D.
Departments of Physiological Science and Molecular Biology, Fukuoka Dental College, Fukuoka 814-0193, Japan
Email: hatta@college.fdcnet.ac.jp